**Author's response to reviews**

**Title:** Dengue hemorrhagic fever and severe thrombocytopenia in a patient on mandatory anticoagulation; balancing two life threatening conditions; a case report

**Authors:**

Champika SSK Gamakaranage (champikasri@gmail.com)
Chaturaka Rodrigo (chaturaka.rodrigo@gmail.com)
Sincy Samarawickrama (ayodya.rodrigo@gmail.com)
Dilushi Wijayaratne (dilushi_w@yahoo.com)
Malaka Jayawardane (djmalaka@gmail.com)
Panduka Karunanayake (pandukaru@hotmail.com)
Saroj Jayasinghe (sarojoffice@yahoo.com)

**Version:** 2 **Date:** 13 April 2012

**Author's response to reviews:**

Dear editor,

Re: query on consent

We apologize for the mistake. The consent was obtained from the patient herself. The relatives consent was obtained for a different patient whose case report was sent to J Med Case reports. The wrong sentence had been copied on to this manuscript. It is now corrected.

Kind regards
Chaturaka Rodrigo